\begin{table}%T3
\par
\caption{\label{t:ang_diam}\label{tab:theta}Stars with measured interferometric angular diameters.}
\small%\centering
\par
\begin{tabular}{rcccrc} \hline\hline\noalign{\smallskip}
HD~ & $\theta_{\rm LD}$ & Ref.\tablefootmark{a} & $\teff^{{\rm IRFM}}$ & $\feh$ & $\theta_{\rm IRFM}$ \\  
   & mas & & K & dex & mas \\ \hline
  3651 & $0.790\pm0.027 $ & 1 & 5234 & $ 0.15$ & $0.756\pm0.022$ \\
  6582 & $0.973\pm0.009 $ & 2 & 5403 & $-0.84$ & $0.954\pm0.021$ \\
  9826 & $1.114\pm0.009 $ & 1 & 6151 & $ 0.10$ & $1.121\pm0.023$ \\
 10700 & $2.078\pm0.031 $ & 3 & 5364 & $-0.53$ & $2.089\pm0.026$ \\
 10780 & $0.763\pm0.021 $ & 2 & 5317 & $ 0.01$ & $0.806\pm0.022$ \\
 19994 & $0.788\pm0.026 $ & 1 & 6020 & $ 0.18$ & $0.746\pm0.009$ \\
 22049 & $2.148\pm0.029 $ & 3 & 5056 & $-0.09$ & $2.200\pm0.032$ \\
 23249 & $2.394\pm0.029 $ & 3 & 5060 & $ 0.08$ & $2.399\pm0.059$ \\
 26965 & $1.650\pm0.060 $ & 3 & 5188 & $-0.27$ & $1.482\pm0.018$ \\
 61421 & $5.443\pm0.030 $ & 3 & 6626 & $ 0.00$ & $5.326\pm0.068$ \\
 75732 & $0.854\pm0.024 $ & 1 & 5282 & $ 0.38$ & $0.718\pm0.025$ \\
102870 & $1.450\pm0.018 $ & 4 & 6100 & $ 0.13$ & $1.426\pm0.014$ \\
117176 & $1.009\pm0.024 $ & 1 & 5540 & $-0.06$ & $0.969\pm0.021$ \\
120136 & $0.786\pm0.016 $ & 1 & 6407 & $ 0.28$ & $0.840\pm0.019$ \\
121370 & $2.244\pm0.019 $ & 3 & 6052 & $ 0.26$ & $2.214\pm0.043$ \\
128620 & $8.511\pm0.020 $ & 3 & 5772 & $ 0.20$ & $8.511\pm0.079$ \\
128621 & $6.000\pm0.021 $ & 5 & 5217 & $ 0.23$ & $6.151\pm0.234$ \\
131977 & $1.230\pm0.030 $ & 3 & 4633 & $ 0.04$ & $1.162\pm0.054$ \\
150680 & $2.397\pm0.044 $ & 3 & 5780 & $ 0.03$ & $2.352\pm0.055$ \\
161797 & $1.953\pm0.039 $ & 3 & 5520 & $ 0.22$ & $2.004\pm0.050$ \\
185144 & $1.254\pm0.012 $ & 2 & 5293 & $-0.21$ & $1.261\pm0.029$ \\
188512 & $2.180\pm0.090 $ & 6 & 5164 & $-0.18$ & $2.070\pm0.049$ \\
190360 & $0.698\pm0.019 $ & 1 & 5564 & $ 0.21$ & $0.673\pm0.017$ \\
198149 & $2.650\pm0.040 $ & 3 & 4980 & $-0.16$ & $2.720\pm0.090$ \\
201091 & $1.775\pm0.013 $ & 3 & 4429 & $-0.24$ & $1.706\pm0.070$ \\
209100 & $1.890\pm0.020 $ & 3 & 4665 & $-0.06$ & $1.825\pm0.021$ \\
217014 & $0.748\pm0.027 $ & 1 & 5754 & $ 0.17$ & $0.698\pm0.019$ \\
\hline
\end{tabular}
\tablefoot{\tablefoottext{a}{1. \citet{baines08}; 2. \citet{boyajian08};
  3.  \citet{kervella08} (weighted average if more than one
  measurement was available); 4. \citet{north09}; 5.  \citet{bigot06};
  6.  \citet{nordgren99}.}}
\end{table}